

Systemic inflammation increases across distinct stages of advanced chronic liver disease and correlates with decompensation and mortality

Dalila Costa^{1,2,3,4,†}, Benedikt Simbrunner^{3,4,5,6,7,†}, Mathias Jachs^{3,4,5}, Lukas Hartl^{3,4,5}, David Bauer^{3,4,5,6,7}, Rafael Paternostro^{3,4}, Philipp Schwabl^{3,4,5}, Bernhard Scheiner^{3,4}, Albert Friedrich Stättermayer³, Matthias Pinter³, Michael Trauner³, Mattias Mandorfer^{3,4,5}, Thomas Reiberger^{3,4,5,6,7,*}

¹Life and Health Sciences Research Institute (ICVS), School of Medicine, University of Minho, Braga, Portugal; ²Gastroenterology Department, Braga Hospital, Braga, Portugal; ³Division of Gastroenterology and Hepatology, Department of Medicine III, Medical University of Vienna, Vienna, Austria; ⁴Vienna Hepatic Hemodynamic Laboratory, Medical University of Vienna, Vienna, Austria; ⁵Christian Doppler Laboratory for Portal Hypertension and Liver Fibrosis, Medical University of Vienna, Vienna, Austria; ⁶Ludwig Boltzmann Institute for Rare and Undiagnosed Diseases (LBI-RUD), Vienna, Austria; ⁷CeMM Research Center for Molecular Medicine of the Austrian Academy of Sciences, Vienna, Austria

Background & Aims: Distinct prognostic stages of advanced chronic liver disease (ACLD) are defined by severity of portal hypertension (PH) and the presence/absence of clinical complications. We characterised the degree of liver dysfunction, PH, and systemic inflammation across the distinct prognostic stages and assessed their relative impact on decompensation and mortality.

Methods: A single-centre, prospective cohort of ACLD patients undergoing hepatic venous pressure gradient (HVPG) measurement between 01/2017 and 08/2019 were classified into 6 prognostic stages: mild PH (HVPG 6–9 mmHg, S0), clinically significant PH (HVPG ≥10 mmHg without varices, S1), presence of varices (S2), history of variceal bleeding (S3), first nonbleeding decompensation (S4), and further decompensation (S5). The model for end-stage liver disease (MELD), C-reactive protein (CRP), and IL-6 levels were assessed in relation to their predictive value for decompensation and death.

Results: Among 168 ACLD patients 78 had compensated (cACLD, S0 = 13; S1 = 21; S2 = 44) and 90 had decompensated (dACLD, S3 = 10; S4 = 58; S5 = 22) disease. MELD increased across all stages (p < 0.001), whereas HVPG mostly increased within cACLD substages. Significant increases in CRP and IL-6 levels were only noted across dACLD substages. IL-6 was an independent predictor of decompensation at 1-year follow-up in cACLD (hazard ratio [HR] 1.06, 95% CI 1.01–1.10; p = 0.013). In dACLD patients, IL-6 levels predicted death/transplantation after 1-year of follow-up (HR 1.02, 95% CI 1.01–1.03; p = 0.004).

[†] These authors contributed equally to this study and share the first authorship. https://doi.org/10.1016/j.jhep.2020.10.004



Conclusion: HVPG progression occurs mostly in cACLD patients, whereas systemic inflammation, as reflected by IL-6 levels, only increases substantially across dACLD stages. IL-6 levels correlate with the risk of first decompensation in cACLD and of death/ transplantation in dACLD patients.

Lay summary: Patients with advanced chronic liver disease (ACLD; *i.e.* liver cirrhosis) have a certain risk of mortality according to their stage of disease. Progression of disease is greatly influenced by increased pressure in the portal venous system (*i.e.* portal hypertension) and occurrence of clinical complications (*i.e.* decompensation). Our study demonstrates that systemic inflammation markedly increases across highest disease stages, and the inflammation biomarker IL-6 in blood may specifically indicate risk of disease progression in patients with ACLD.

Clinical Trials registration: The study is registered at ClinicalTrials.gov (NCT03267615).

© 2020 European Association for the Study of the Liver. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Introduction

The clinical course of advanced chronic liver disease (ACLD) is characterised by a transition from a clinically compensated state (cACLD) to a symptomatic decompensated state (dACLD) which indicates a considerably increased risk for mortality.^{1,2} Portal hypertension (PH) represents a key driver of decompensation and mortality in ACLD patients.³ Moreover, patients with ACLD are at risk of developing acute-on-chronic liver failure (ACLF), a systemic inflammatory syndrome associated with multi-organ dysfunction.^{4,5} Considering the different risks of disease progression and mortality, growing evidence suggests that the natural history of cirrhosis, i.e. ACLD, should be described by a multistate model.^{2,6} The Baveno IV consensus conference⁷ redefined cirrhosis in 4 substages, however, more recently, compensated cirrhosis has been subdivided into mild portal hypertension (hepatic venous pressure gradient [HVPG] of 6-9 mmHg), clinically significant PH (CSPH; HVPG ≥10 mmHg but without varices), and presence of gastroesophageal varices



Keywords: Cirrhosis; Portal hypertension; Systemic inflammation; Interleukin-6; Creactive protein.

Received 4 August 2020; received in revised form 18 September 2020; accepted 7 October 2020; available online 16 October 2020

^{*} Corresponding author. Address: Division of Gastroenterology and Hepatology, Department of Internal Medicine III, Medical University of Vienna, Waehringer Guertel 18-20, A-1090, Vienna, Austria. Tel.: +43-1-40400-47410; Fax: +43-1-40400-47350.

E-mail address: thomas.reiberger@meduniwien.ac.at (T. Reiberger).

(GEV).^{2,8} Three substages with increasing risk of mortality have been proposed for dACLD, defined by the occurrence of variceal bleeding alone, any first non-bleeding decompensating event (mostly frequently ascites) and any further decompensating event.²

This multistate model enables a prognostic classification of patients regarding their distinct risks for different clinical outcomes.^{2,6} The Child-Pugh⁹ and model for end-stage liver disease (MELD)^{10,11} scores are currently used to define prognosis by assessing liver dysfunction, but do not necessarily reflect the dynamic clinical state of patients with cirrhosis.⁶ Measurement of HVPG provides an accurate assessment of portal pressure and is the best predictor of the development of GEV¹² and decompensation.³ In addition to the severity of PH, systemic hemodynamic derangements play an important role in the development of further decompensating events such as refractory ascites and hepatorenal syndrome (HRS).^{3,6,13} More recently, is has been recognised that systemic inflammation (SI) increases with severity of PH and circulatory dysfunction and may be a key driver of clinical detoriation.^{14,15} Abnormal translocation of bacteria and pathogen-associated molecular patterns from the intestinal lumen into the portal and systemic circulation is considered the main pathophysiological mechanism of SI in cirrhosis,^{15,16} highlighting the central role of the gut–liver axis.¹⁷ Although acute decompensation may be triggered by significant SI, the exact chronological relationship between bacterial translocation, SI, and decompensation currently remains elusive.¹⁶

A better understanding of the pathophysiological factors driving the transition across ACLD stages is necessary to redefine therapeutic strategies. To this end, a comprehensive characterisation of the interplay between hepatic dysfunction, severity of PH, and SI is required. Thus, this study was designed (i) to characterise the progression of liver dysfunction, PH and SI across distinct prognostic clinical stages of ACLD and (ii) to evaluate their relative impact on the risk for decompensation and mortality.

Patients and methods

Study design and patients

We performed a prospective, observational, single-centre cohort study in consecutively recruited patients with ACLD undergoing HVPG measurement at the Vienna General Hospital of the Medical University of Vienna between January 2017 and August 2019. Inclusion criteria were (i) confirmed diagnosis of ACLD (based on clinical, biochemical, imaging, and/or histological criteria) and (ii) HVPG >5 mmHg confirming the presence of PH. Patients meeting any of the following criteria were excluded: age <18 years, HVPG measurement while on non-selective betablockers (NSBBs), ACLF, active bacterial infection at evaluation, hepatocellular carcinoma, active alcohol abuse, previous transjugular intrahepatic portosystemic shunt (TIPS) insertion, occlusive portal vein thrombosis, or liver transplantation (LT), and chronic kidney disease. The study was conducted in accordance with the principles of the Declaration of Helsinki, was approved by the local ethics committee and registered at www. clinicaltrials.org (NCT03267615). All patients gave written informed consent for participation in this study.

Clinical stages of ACLD

Patients were classified according to the recently defined clinical prognostic stages, adapted from D'Amico et al.² Compensated ACLD (cACLD) was defined as absence of any decompensation event and cACLD was further divided into 3 substages: stage (S) SO: subclinical PH with HVPG 6–9 mmHg; S1: CSPH with HVPG ≥10 mmHg without GEV; S2: presence of GEV. Decompensated ACLD (dACLD) was defined by the presence or history of at least 1 decompensating event, that is ascites, variceal bleeding, and overt hepatic encephalopathy (HE). Importantly, overt HE was not observed in the absence of other decompensating events in our cohort. Moreover, as diagnosis of covert HE is challenging and its prognostic value remains to be defined,² covert HE was not considered as a decompensation event in this study. Patients with dACLD were subclassified into 3 substages: S3: history of acute variceal bleeding; S4: first non-bleeding decompensation (mostly frequently ascites); S5: further decompensation as defined by either ascites plus bleeding, refractory ascites according to International Ascites Club criteria,¹⁸ HRS, or spontaneous bacterial peritonitis (SBP).

HVPG measurement

HVPG measurement was performed within this study to establish the diagnosis of CSPH for prognostication as supported by current guidelines.^{19,20} Briefly, under local anaesthesia and ultrasound guidance, a catheter introducer sheath was inserted into the right internal jugular vein. Subsequently, a hepatic vein was cannulated, and the free and hepatic venous pressures were obtained at least as triplicate measurements using a balloon catheter,²¹ according to a standardised protocol, as previously described in detail.²²

Clinical and laboratory parameters

In addition to routine laboratory tests used to compute MELD and Child-Pugh scores as measures of liver dysfunction severity, serum levels of C-reactive protein (CRP) and interleukin-6 (IL-6) were assessed as biomarkers of SI at the time of liver vein catheterisation. Determination of CRP (upper limit of normal, ULN <0.5 mg/dl) and of IL-6 (ULN <7 ng/dl) was performed at the ISO-certified laboratory of the Vienna General Hospital following the manufacturer's instructions.

Clinical follow-up for outcomes and mortality

Patients were followed after HVPG until their last clinical visit, LT, or death. First decompensation was the main outcome parameter of interest in patients with cACLD (stages S0–S2), whereas liver-related mortality/LT was considered the main outcome parameter of interest for patients with dACLD (stages S3–S5).

Statistical analysis

All statistical analyses were performed using IBM SPSS Statistics 25 (IBM, New York, NY, USA) or GraphPad Prism 8 (GraphPad Software, CA, USA). Categorical variables were reported as absolute frequencies (n) and relative frequencies (%); continuous variables as mean \pm SD or median with IQR, as appropriate. Normal distribution was assessed by Kolmogorov-Smirnov and Shapiro-Wilk tests. Kruskal-Wallis test was used to compare



Fig. 1. Patient flow chart. ACLD, advanced chronic liver disease; ACLF, acute-on-chronic liver failure; cADLD, compensated ACLD; dACLD, decompensated ACLD; HVPG, hepatic venous pressure gradient; NSBB, Non-selective beta-blockers; Q, quartile.

non-parametric continuous variables and when significant differences were attained, Mann-Whitney U test with Bonferroni correction was applied. One-way ANOVA with Tukey *post-hoc* test was used to compare parametric continuous variables. The analyses of clinical outcomes (decompensation and liver-related mortality/transplantation) were performed separately for cACLD and dACLD patients. The time-dependent incidence rate of events of interests was obtained from Kaplan-Meier estimates and comparison of cumulative incidence curves was performed by the log-rank test. Multivariable analysis of independent predictors of the event of interest was performed by a backward stepwise Cox proportional hazards regression model considering the clinical outcome 1 year after HVPG measurement. The level of significance was set at 2-sided *p* value <0.05.

Results

Patients characteristics

During the study period, 321 patients with ACLD underwent HVPG measurement, of whom 168 fulfilled the inclusion and no exclusion criteria and were finally included in this study (Fig. 1 and Table 1). The median age was 57.2 years and most patients were male (n = 114, 67.9%). The main aetiologies were alcoholrelated liver disease (n = 77, 45.8%) and viral hepatitis (n = 36, 21.4%). At baseline, 78 (46.4%) patients were compensated with 13 (16.7%) in substage S0 (HVPG 6-9 mmHg), 21 (26.9%) in S1 (HVPG \geq 10 mmHg without GEV), and 44 (56.4%) in S2 with GEV. Representative liver biopsy specimens were available in 59 (35.1%) patients for histological confirmation of cirrhosis. Among patients with dACLD (n = 90, 53.6%), 10 (11.1%) had experienced variceal bleeding (S3); 58 (64.4%) had ascites (S4), and 22 (24.4%) had further decompensation (S5). Within decompensated substage S5, 7 patients had ascites and bleeding, 11 patients had refractory ascites, 6 had hepatorenal syndrome, 3 had SBP (some patients had more than 1 further decompensation event).

Progressive hepatic dysfunction was observed by increasing MELD across the distinct substages (median 9 in S0 to 14 in S5),

with a significant difference noted in the transition from cACLD to S4 and to S5 (*p* <0.001; Fig. 2). Naturally, the Child-Pugh score significantly differentiated compensated from decompensated ACLD, however, it had a lower discriminative ability to distinguish the clinical substages (Table S1). HVPG progressively increased across clinical substages with most pronounced increases within the cACLD substages (median 8 mmHg in S0 to 17.5 mmHg in S2) and no significant difference was observed between S4 and S5 (Fig. 2). Regarding SI, both CRP and IL-6 increased only across the dACLD substages with IL-6 levels being significantly higher in S4 (median 10.6 pg/ml) and S5 (median 28.2 pg/ml) as compared with S0 (median 4.7 pg/ml) and to S2 (median 5.7 pg/ml; Fig. 2). The proportion of patients that showed IL-6 and CRP levels within the normal range were 70.0% and 83.3%, respectively, in cACLD, but only 29.3% and 46.7% of dACLD patients, respectively, had normal IL-6 or CRP values.

Interestingly, the degree of SI in patients with a history of variceal bleeding (S3: IL-6 median 5.7 pg/ml) was similar to the patients with cACLD (S0–S2: IL-6 median 5.8 pg/ml; p = 0.825) and SI was also significantly less pronounced as compared to the "next" dACLD substages (S4: IL-6 median 10.6 pg/ml; p = 0.009) and to the subgroup of dACLD S5 patients with a history of variceal bleeding (IL-6 median 12.3 pg/ml; 0.034). The time periods from previous variceal bleeding to the assessment of SI (*i.e.* IL-6 level determination) were a median of 18.9 months in S3 and 16.2 months in S5. Notably, there were no significant differences in IL-6 values between this subgroup of S5 patients and S4, p = 0.519 (Fig. S1). White blood cell (WBC) counts were similar between patients with cACLD and dACLD, p = 0.242 (Fig. S2).

Follow-up events and clinical outcomes

The median follow-up of the included study cohort was 12.2 months. The number of events per clinical state are displayed in Table 1 and Table S1. Among cACLD patients, 1 (1.3%) patient transitioned to S3, 8 (10.3%) patients to S4 and 1 patient (1.3%) to S5. Overall, 5 (6.4%) patients died, liver-related death was

recorded in 3 (3.8%) of patients. Among the dACLD patients, 2 (20%) patients in S3 experienced variceal re-bleeding and 4 (40%) patients progressed to substage S5 as a result of further decompensation. In substage S4 patients, 3 patients underwent LT and 10 (17.2%) patients progressed to S5 (*i.e.* experienced further decompensation. In S5, 4 (18.8%) patients underwent LT. The liver-related death rates were 10%, 10.3%, and 22.7% in the substage S3, S4, and S5, respectively.

First decompensation in cACLD

The 12-month probability of developing a decompensation event was 0%, 6%, and 16%, for substages S0, S1, and S2, respectively (Fig. 3A). Despite numerically higher risk for decompensation in

stage S2 patients with varices, the difference did not attain statistical significance (p = 0.297). cACLD with elevated IL-6 ($\geq 7 \text{ pg}/\text{ml}$) or CRP ($\geq 0.5 \text{ mg/dl}$) levels did only show a non-significantly increased probability of developing a first decompensation event (Fig. 3B and C). Importantly, after adjusting for covariables using a multivariate Cox proportional hazard regression model IL-6 levels were an independent predictor of decompensation at 1year follow-up (hazard ratio, HR: 1.06; 95% CI: 1.01–1.10; p =0.013) with MELD showing a strong trend towards significance (HR 1.04; 95% CI: 1.00–1.62; p = 0.054; see Model 1 in Table 2). When including CRP instead of IL-6 using into the multivariate Cox regression, CRP (HR 1.90; 95% CI 0.97–3.74; p = 0.062) and HVPG (HR 1.13; 95% CI 1–1.28; p = 0.059) had a strong but non-

Table 1. Patient characteristics at baseline HVPG measure	ment and follow-up events in the 6 substages of ACLD.
---	---

	Compensated (n = 78 patients, 46.4%) Decompensated (n = 90 patients, 52		Decompensated (n = 90 patients, 53.6%)				
	Stage 0	Stage 1	Stage 2	Stage 3	Stage 4	Stage 5	p value
Definition	HVPG 6-9	HVPG ≥10	GEV	Bleeding	Ascites	Further decompensation ^a	
Patients, n	13	21	44	10	58	22	
Age (years)	52.3 (16.0)	55.4 (17.0)	58.8 (16.0)	52.7 (8.0)	57.2 (16.0)	60.5 (14.0)	0.142
Sex; male; n (%)	10 (76.9)	11 (52.4)	31 (70.5)	7 (70.0)	38 (65.5)	17 (77,3)	0.550
Aetiology; n (%)							
ALD	5 (38.5)	2 (9.5)	9 (20.5)	6 (60.0)	42 (72.4)	13 (59.1)	
VIRAL	4 (30.8)	9 (42.9)	14 (31.8)	3 (30.0)	3 (5.2)	3 (13.6)	
ALD+VIRAL	0 (0.0)	4 (19.0)	3 (6.8)	0 (0.0)	7 (12.1)	2 (9.1)	
NASH	1 (7.7)	3 (14.3)	11 (25.0)	0 (0.0)	1 (1.7)	2 (9.1)	
CHOL	0 (0.0)	2 (9.5)	2 (4.5)	0 (0.0)	0 (0.0)	0 (0.0)	
OTHER	3 (23.1)	1 (4.8)	4 (9.1)	1 (10.0)	5 (8.6)	2 (9.1)	
GEV; n (%)					· · ·		
Small GEV	0 (0.0)	0 (0)	25 (56.8)	0 (0.0)	18 (31.0)	7 (31.8)	
Large GEV	0 (0.0)	0 (0)	19 (43.2)	10 (100.0)	20 (34.5)	9 (40.9)	
Child-Pugh score	5 (0.0)	5 (0.0)	5 (1.0)	5 (1.0)	8 (2.0)*†‡§	8 (2.0)*†‡§	<0.001
MELD	9 (5.0)	9 (2.0)	10 (4.0)	11 (3.0)	13 (6.0)*†‡	14 (4.0)*†‡	<0.001
MELD-Na; x (SE)	8.9 (3.0)	10.0 (2.8)	10.6 (3.6)	11.1 (2.5)	16.2 (4.8)*†‡§	16.5 (4.6)* ^{†‡§}	<0.001
HVPG (mmHg)	8 (3.0)	13 (3.0)*	17.5 (9.0)*	16.5 (5.0)*†	19.5 (7.0)*†‡	20 (8.0)*†	<0.001
Heart rate (per min)	76 (25.0)	82 (26.0)	72 (17.0)	68.5 (7.0)	80 (26.0)	81 (24.0)	0.119
MAP (mmHg)	102 (22.0)	98 (26.0)	109 (17.0)	98.5 (22.0)	98 (19.0)	96 (19.0)	0.082
Hb (g/dl)	13.9 (2.2)	13.3 (1.5)	12.2 (2.3)*†	11.4 (2.8)*†‡	11.3 (2.2)*†	9.7 (3.1)* ^{††¶}	<0.001
PLT (G/L)	106 (56.0)	127 (62,8)	81.5 (33)* [†]	63 (77.5)* [†]	107 (62.8)	107.5 (119.3) ^{‡§}	<0.001
WBC (G/L)	5.2 (3.2)	6.7 (2.9)	3.5 (2.1)*†	3.2 (3.3)*†	4.9 (2.5) ^{‡§}	4.3 (2.9)	<0.001
Creatinine (mg/dl)	0.8 (0.3)	0.7 (0.3)	0.7 (0.3)	0.8 (0.5)	0.7 (0.3)	1.0 (1.0)	0.116
Sodium (mmol/L)	140 (3.0)	140 (3.0)	140 (4.0)	140.5 (4.3)	136 (4.0)* ^{†‡§}	137 (9.3) ^{>}	<0.001
Bilirubin (mg/dl)	0.8 (0.4)	1.0 (0.8)	0.9 (0.8)	1.0 (0.6)	1.6 (1.8)*†‡	1.1 (1.7)	0.001
Albumin (g/L)	41.9 (5.4)	39.8 (4.3)	39.7 (6.2)	39.7 (4.8)	33.8 (7.0)* ^{†‡§}	32.4 (6.8)* ^{†‡§}	<0.001
INR	1.2 (0.2)	1.2 (0.3)	1.3 (0.3)	1.4 (0.2)	1.5 (0.3)* ^{†‡}	1.5 (0.4)	<0.001
CRP (mg/dl)	0.1 (0.2)	0.2 (0.3)	0.2 (0.3)	0.1 (0.1)	0.5 (0.9)* ^{†‡§}	1.0 (2.4) ^{*†‡§}	<0.001
IL-6 (pg/ml)	4.7 (2.7)	7.0 (9.2)	5.7 (4.6)	5.7 (6.9)	10.6 (14.9)* [‡]	28.2(42.8)* ^{†‡§} ¶	<0.001
LBP (µg/ml)	7.2 (2.7)	6.6 (2.3)	6.5 (2.5)	5.7 (2.0)	6.8 (4.4)	8.9 (6.1)	0.052
PCT (ng/ml)	0.04 (0.06)	0.08 (0.07)	0.07 (0.06)	0.05 (0.1)	0.1 (0.1)*	0.1 (0.1)*	0.001
FU event ^b , n (%)	0 (0)	2 (9.5%)	9 (20.5%)	6 (60%)	13 (22.4%)	11 (50.0%)	
Decompensation ^c	-	2 (9.5%)	9 (20.5%)	6 (60%)	10 (17.2%)	7 (31.8%)	
Liver-related death	-	1 (4.8%)	2 (4.5%)	1 (10%)	6 (10.3%)	5 (22.7%)	
LT	-	-	_	1 (10%)	4 (6.9%)	4 (18.2%)	

Unless indicated otherwise, metric variables are presented as median (IQR).

ALD, alcohol-related liver disease; CHOL, cholestastic liver disease; CRP, C-reactive protein; FU, follow-up; GEV, gastroesophageal varices; Hb, haemoglobin; HR, heart rate; HVPG, hepatic venous pressure gradient; INR, international normalised ratio; IL-6, interleukin-6; LBP, lipopolysaccharide binding protein; LT, orthotopic liver transplantation; MAP, mean arterial pressure; MELD, model for end-stage liver disease; MELD-Na, MELD including serum sodium; NASH, non-alcoholic steatohepatitis; PCT, procalcitonin; PLT, platelet count; WBC, white blood cell count.

Statistical analysis: Kruskal-Wallis and one-way ANOVA were used to compare non-parametric and parametric variables, respectively.

^aStage 5 includes patients with further decompensation due to ascites and bleeding, refractory ascites, hepatorenal syndrome, or spontaneous bacterial peritonitis. ^bMedian follow-up duration was 12.2 months (max. 36.8 months).

^cConsidering first decompensation in cACLD patients, and further decompensation in dACLD patients.

*p <0.05 when compared to stage 0.

[†]vs. stage 1.

[‡]vs. stage 2.

[§]vs. stage 3.

¶vs. stage 4.

JOURNAL OF HEPATOLOGY

significant independent predictive value for first decompensation in cACLD (see Model 2 in Table 2).

Predictors of mortality and/or LT in dACLD

The 3 substages within dACLD patients were associated with a progressively increasing probability of liver-related death/LT after 1 year of 0%, 19%, and 48%, for substages S3, S4, and S5, respectively (Fig. 4A; p < 0.001). Decompensated patients with high IL-6 (\geq 14 pg/ml, log-rank p < 0.001; Fig. 4B) and CRP (\geq 0.5 mg/dl; log-rank p = 0.016; Fig. 4C) showed a significantly higher probability of liver-related death/LT. The Cox proportional hazard regression analysis identified IL-6 levels (HR 1.02; 95% CI 1.01–1.03; p = 0.004; see Model 1 in Table 3) and CRP levels (HR 1.51; 95% CI 1.16–1.98; p = 0.003; see Model 2 in Table 3) as independent predictors of death/LT at 1-year follow-up in dACLD patients.

Discussion

This study shows the interplay between liver function, PH and SI throughout distinct prognostic stages of ACLD. Next to the detailed clinical characterisation of our ACLD cohort, we used HVPG as the diagnostic gold-standard for assessing severity of PH, and determination of IL-6 and CRP levels as established biomarkers of SI to refine the characterisation of the complex

clinical multistate ACLD model. As the outcomes and underlying pathophysiological drivers of disease procession are considerably different between patients with compensated and decompensated ACLD, prognostic indicators may be distinct or may have different weight across distinct ACLD substages.^{1,2} Of note, patients in this study were electively referred to liver vein catheterisation and, thus, may not be fully representative of patients with ACLD treated at our institution, which constitutes a limitation of this study. Consequently, as patients with acute decompensation (i.e. nonelective admission) at the time of HVPG measurement were not included, patients were in a stable disease state to be eligible for this study. Therefore, the levels of SI in our study might not be similar to those observed in patients when sampled at the time of acute decompensation and/or pre-ACLF, as performed in the PREDICT study.²¹

In cACLD patients, a small but steady increase of MELD across indicative of progressive hepatic dysfunction was noted. However, most pronounced increases were noted in severity of PH as evident by considerable increases in HVPG across the different substages of cACLD. These results are in line with previous studies suggesting HVPG as the best predictor for the development of varices,¹² that is progression to substage S2. Most patients in compensated stages displayed normal values of IL-6 and



Fig. 2. Degree of hepatic dysfunction, portal pressure and systemic inflammation across ACLD substages. (A) Dynamics of median MELD, Child-Pugh score, HVPG, CRP and IL-6 serum levels across ACLD substages. (B–E) Median MELD, HVPG, CRP and IL-6 serum levels across ACLD substages (numbers indicate median levels and bars represent IQR). ACLD, advanced chronic liver disease; cACLD, compensated ACLD; CRP, C-reactive protein; dACLD, decompensated ACLD; HVPG, hepatic venous pressure gradient; IL-6, interleukin-6; MELD, model for end-stage liver disease.



Fig. 3. Rates of first decompensation in compensated ACLD. Patients with cACLD stratified by (A) clinical substages (S0–S2), (B) IL-6 levels, and (C) CRP levels. Incidence rate of first decompensation was obtained from Kaplan-Meier estimates and compared using log-rank test (level of significance *p* <0.05). ACLD, advanced chronic liver disease; cACLD, compensated ACLD; CRP, C-reactive protein; IL-6, interleukin-6.

CRP, and there were no significant increases across the substages S0–S2 among cACLD patients. Recently, Turco *et al.*¹³ showed a progressively higher proportion of patients with SI as defined by elevated CRP levels in the 3 compensated substages. However, other clinical and experimental studies would support the observation of increasing levels of proinflammatory cytokines and worsening of splanchnic vasodilation predominantly limited to dACLD patients with ascites.^{24,25}

We observed a numerically increasing risk for decompensation across the 3 cACLD substages, supporting the key prognostic role CSPH (*i.e.* substage S1) to predict development of hepatic decompensation.^{3,12} Similarly, the presence of GEV (*i.e.* cACLD substage S2) increases the risk of variceal bleeding, nonbleeding decompensation and death.^{8,20} Importantly, we observed that the transition across the distinct prognostic substages is not essentially sequential, but instead cACLD patients may shift to any of the dACLD substages, most frequently to S4 because of the development of ascites as the most common first manifestation of hepatic decompensation.²⁶ Although we observed a clear trend for patients with abnormal IL-6 levels towards higher risk of first decompensation, this was not evident for CRP levels, which may be a less sensitive or less specific biomarker of SI. Notably, after adjusting for prognostic covariables, IL-6 remained an independent predictor of first decompensation in our cohort. When IL-6 levels were not considered in the multivariate regression model, both CRP

Table 2. Cox proportional hazard regression model assessing potential predictors of decompensation after 1 year in compensated ACLD patients (substages S0–S2).

		Univariable			Multivariable	
Model 1	HR	95% CI	p value	HR	95% CI	p value
Age (per year)	1.04	0.97-1.11	0.285			
MELD (per point)	1.26	1.02-1.55	0.032	1.27	1.00-1.62	0.054
HVPG (per mmHg)	1.13	0.99-1.30	0.068	1.04	0.89-1.22	0.622
Albumin (g/L)	0.90	0.79-1.02	0.100	0.98	0.82-1.18	0.832
WBC (G/L)	0.99	0.86-1.15	0.955			
IL-6 (pg/ml)	1.07	1.02-1.12	0.004	1.06	1.01–1.10	0.013
		Univariable			Multivariable	
Model 2	HR	Univariable 95% CI	p value	HR	Multivariable 95% CI	p value
Model 2 Age (year)	HR 1.04	Univariable 95% CI 0.97-1.11	p value 0.285	HR	Multivariable 95% Cl	p value
Model 2 Age (year) MELD (point)	HR 1.04 1.26	Univariable 95% CI 0.97-1.11 1.02-1.55	<i>p</i> value 0.285 0.032	HR 1.12	Multivariable 95% CI 0.85-1.48	p value 0.418
Model 2 Age (year) MELD (point) HVPG (mmHg)	HR 1.04 1.26 1.13	Univariable 95% CI 0.97-1.11 1.02-1.55 0.99-1.30	<i>p</i> value 0.285 0.032 0.068	HR 1.12 1.13	Multivariable 95% CI 0.85–1.48 1.00–1.28	p value 0.418 0.059
Model 2 Age (year) MELD (point) HVPG (mmHg) Albumin (g/L)	HR 1.04 1.26 1.13 0.90	Univariable 95% CI 0.97-1.11 1.02-1.55 0.99-1.30 0.79-1.02	<i>p</i> value 0.285 0.032 0.068 0.100	HR 1.12 1.13 0.93	Multivariable 95% CI 0.85–1.48 1.00–1.28 0.80–1.08	<i>p</i> value 0.418 0.059 0.331
Model 2 Age (year) MELD (point) HVPG (mmHg) Albumin (g/L) WBC (G/L)	HR 1.04 1.26 1.13 0.90 0.99	Univariable 95% CI 0.97-1.11 1.02-1.55 0.99-1.30 0.79-1.02 0.86-1.15	<i>p</i> value 0.285 0.032 0.068 0.100 0.955	HR 1.12 1.13 0.93	Multivariable 95% CI 0.85–1.48 1.00–1.28 0.80–1.08	<i>p</i> value 0.418 0.059 0.331

Multivariable analysis was performed using a backward stepwise Cox proportional hazards regression model.

ACLD, advanced chronic liver disease; CRP, C-reactive protein; HVPG, hepatic venous pressure gradient; HR, hazard ratio; IL-6, interleukin-6; MELD, model for end-stage liver disease; WBC, white blood cells.

levels and HVPG had a strong, but non-significant predictive value for first decompensation in cACLD patients. Although the prognostic role of HVPG and CRP for first decompensation has been previously reported,^{3,27,28} our study is the first to demonstrate the predictive power of IL-6 for first decompensation in cACLD patients.

IL-6 is a pleiotropic proinflammatory cytokine that is released by activated monocytes and macrophages activation, for example upon Toll-like receptor 4 stimulation. Subsequently, IL-6 induces an acute phase response, that may be paralleled by a CRP release in the liver.^{5,16,29} Our results suggest that IL-6 is a more sensitive predictor of decompensation and represent a valuable biomarker for SI in cirrhosis. Even though these data should be further explored, our results reinforce the SI hypothesis,¹⁶ which states that bacterial translocation and SI are key drivers of hepatic decompensation.

In dACLD patients, MELD continued to increase across the decompensated substages with most pronounced increases noted in patients with ascites (*i.e.* substage S4). Importantly, CRP and IL-6 levels considerably increased across the distinct substages of dACLD, with incremental increases of IL-6 levels noted in patients with ascites and with further decompensation, that is in substages S4 and S5.

Of note, the degree of SI in patients with a history of variceal bleeding (S3) did not differ from cACLD patients within substages S0-S2. Furthermore, dACLD patients with ascites and a history of variceal bleeding (a subgroup of S5) displayed the same levels of inflammation markers as patients with only ascites (substage S4). In both scenarios, the median time from previous variceal bleeding to HVPG measurement was longer than 1 year. One could argue that once variceal bleeding resolves, patients return to a re-compensated stage after a certain amount of time, or that the onset of significant SI is only linked to persisting ascites and/or further decompensation events. Recently published data from the PREDICT study identified 3 different courses of acute decompensation that showed different associations with PH and SI as decisive pathomechanisms.²³ In the subgroup of patients presenting with first hepatic decompensation, levels of SI were lowest in patients with gastrointestinal haemorrhage, as compared with patients with ascites or HE. These results are in line with our study, as we observed similar levels of SI in patients in S3 (after variceal bleeding) as compared with compensated patients. However, as our study cohort explicitly excluded patients with non-elective admission at the time of HVPG measurement (*e.g.* acute gastrointestinal haemorrhage), it cannot indicate whether patients with variceal bleeding have increased levels of SI at the timepoint of admission as compared with cACLD patients, or lower levels, as compared with patients with other forms of hepatic decompensation.

These observations evoke potential implications for the definition of re-compensation, which will be discussed during Baveno VII. Even though patients in S3 had a better outcome than the other decompensated substages S4/S5, these patients still experienced a considerable rate of further decompensation during the follow-up. Improved survival rates are consistently reported after acute variceal bleeding,^{30,31} but the subsequent 5-year risk of further decompensation is still as high as 50%.^{14,32} Therefore, it still seems reasonable to differentiate S3 patients from cACLD in terms of prognostication – especially if biomarkers of SI are high. It remains to be assessed in future studies whether the dynamics of IL-6 levels have a role for defining recompensation after an episode of variceal bleeding.

Ascites is the most frequent first non-bleeding decompensating event^{14,33} that defines dACLD substage S4. The significant increase of IL-6 and CRP levels in the transition to S4 and to S5 supports the impact of SI on the natural course of ACLD. Findings from the CANONIC study also revealed that acute decompensation occurs in the setting of SI, which is significantly more severe in patients with ACLF.⁵ Similarly, the PREDICT study suggests that the longitudinal development of SI after acute decompensation greatly determines the clinical course of ACLD.²³ Finally, our results suggest that IL-6 is a strong prognostic marker associated with considerable increased risk of death if not transplanted. Potentially, IL-6 levels might be used in future studies to identify individual



Fig. 4. Incidence of mortality or liver transplantation in decompensated ACLD. Patients with dACLD stratified by (A) clinical substages, (B) IL-6 levels or by (C) CRP levels. Incidence rate of mortality or liver transplantation was obtained from Kaplan-Meier estimates and compared using log-rank test (level of significance *p* <0.05). ACLD, advanced chronic liver disease; CRP, C-reactive protein; dACLD, decompensated ACLD; IL-6, interleukin-6.

dACLD patients who benefit most from therapeutic interventions targeting SI such as albumin administration³⁴ or NSBB intake.^{35,36}

Although we confirmed increased mortality with progressive dACLD substages S3–S5, abnormal levels of CRP and IL-6 also indicated a significant and considerable increased risk for death or transplantation in dACLD patients. Both SI biomarkers IL-6 and CRP were independent predictors of liver-related death/LT, which is supported by previous studies.^{5,13,27,28,37,38} Recently, Remmler *et al.*³² showed that the prognostic value of IL-6 for predicting death after 90 days was even better than of CRP, but this superior predictive value was not observed for death at 1-

year follow-up. Moreover, a recent study by Fernandez *et al.*³⁹ demonstrated substantial fluctuations of IL-6 levels despite an absence of a clear clinical correlate.

Finally, in contrast to reports on the predictive value of WBC counts in acutely hospitalised patients from the CANONIC study^{40,41} we did not observe significant increases of WBC across ACLD stages nor predictive value of WBC for decompensation and mortality. This likely reflects the ability of WBC count to reflect short-term dynamics in patients with acute decompensation and "pre-ACLF" state (*i.e.* patients with non-elective hospitalisation). Our study population rather reflects cACLD and stable dACLD, that is disease stages that are different to "pre-

Table 3. Cox proportional hazard regression model assessing potential predictors of death or liver transplantation after 1 year in decompensated ACLD patients (substages S3–S5).

	Univariable				Multivariable	
Model 1	HR	95% CI	p value	HR	95% CI	p value
Age (per year)	1.05	1.00-1.10	0.075	1.05	1.00-1.11	0.055
MELD (per point)	1.09	0.97-1.22	0.161	1.04	0.89-1.21	0.625
HVPG (per mmHg)	1.09	0.99-1.19	0.072	1.05	0.95-1.17	0.326
Albumin (g/L)	0.89	0.80-0.99	0.027	0.93	0.82-1.06	0.300
WBC (G/L)	0.98	0.80-1.19	0.828			
IL-6 (pg/ml)	1.02	1.01-1.03	0.006	1.02	1.01-1.03	0.004
		Univariable				
		Univariable			Multivariable	
Model 2	HR	Univariable 95% Cl	p value	HR	Multivariable 95% Cl	p value
Model 2 Age (per year)	HR 1.05	Univariable 95% CI 1.00-1.10	p value 0.075	HR 1.05	Multivariable 95% CI 0.99–1.11	p value 0.088
Model 2 Age (per year) MELD (per point)	HR 1.05 1.09	Univariable 95% CI 1.00-1.10 0.97-1.22	<i>p</i> value 0.075 0.161	HR 1.05 1.07	Multivariable 95% CI 0.99–1.11 0.91–1.26	<i>p</i> value 0.088 0.397
Model 2 Age (per year) MELD (per point) HVPG (per mmHg)	HR 1.05 1.09 1.09	Univariable 95% CI 1.00-1.10 0.97-1.22 0.99-1.19	<i>p</i> value 0.075 0.161 0.072	HR 1.05 1.07 1.06	Multivariable 95% CI 0.99–1.11 0.91–1.26 0.96–1.17	<i>p</i> value 0.088 0.397 0.256
Model 2 Age (per year) MELD (per point) HVPG (per mmHg) Albumin (g/L)	HR 1.05 1.09 1.09 0.89	Univariable 95% CI 1.00-1.10 0.97-1.22 0.99-1.19 0.80-0.99	<i>p</i> value 0.075 0.161 0.072 0.027	HR 1.05 1.07 1.06 0.95	Multivariable 95% CI 0.99–1.11 0.91–1.26 0.96–1.17 0.83–1.09	<i>p</i> value 0.088 0.397 0.256 0.465
Model 2 Age (per year) MELD (per point) HVPG (per mmHg) Albumin (g/L) WBC (G/L)	HR 1.05 1.09 1.09 0.89 0.98	Univariable 95% CI 1.00-1.10 0.97-1.22 0.99-1.19 0.80-0.99 0.80-1.19	<i>p</i> value 0.075 0.161 0.072 0.027 0.828	HR 1.05 1.07 1.06 0.95	Multivariable 95% CI 0.99–1.11 0.91–1.26 0.96–1.17 0.83–1.09	<i>p</i> value 0.088 0.397 0.256 0.465

Multivariable analysis was performed using a backward stepwise Cox proportional hazards regression model.

ACLD, advanced chronic liver disease; CRP, C-reactive protein; HVPG, hepatic venous pressure gradient; HR, hazard ratio; IL-6, interleukin-6; MELD, model for end-stage liver disease; WBC, white blood cells.

ACLF" and ACLF which are distinct clinical syndromes.⁴ This hypothesis is also supported by the PREDICT study demonstrating significant decrease of WBC counts after 90 days in patients with a stable course of dACLD.²³

In summary, our study, which assessed thoroughly characterised ACLD patients across distinct prognostic substages, demonstrates that the degree of SI progressively increases with severity of ACLD. Importantly, IL-6 as a biomarker of SI supported the concept of incremental increases in the degree of SI after development of ascites and/or further decompensation. Even if there was no systematic increase in the degree of SI noted among cACLD, IL-6 was an independent predictor of first decompensation. Furthermore, our results suggest IL-6 level as a valuable biomarker of ACLD progression of considerable prognostic relevance, as IL-6 predicted the risk of liver-related death or the need for LT in dACLD patients. If future studies with larger sample size confirm the prognostic value of IL-6 level across distinct ACLD stages, the incorporation of IL-6 into the clinical substage model may facilitate specific therapeutic interventions and personalised medical care for patients with complex and multistage ACLD.

Abbreviations

ACLD, advanced chronic liver disease; ACLF, acute-on-chronic liver failure; c/dACLD, compensated/decompensated advanced chronic liver disease; CRP, C-reactive protein; CSPH, clinically significant portal hypertension; GEV, gastroesophageal varices; HE, hepatic encephalopathy; HRS, hepatorenal syndrome; HVPG, hepatic venous pressure gradient; IL-6, interleukin-6; LT, liver transplantation; MELD, model for end-stage liver disease; NSBB, non-selective beta-blockers; PH, portal hypertension; SBP, spontaneous bacterial peritonitis; SI, systemic inflammation; TIPS, transjugular intrahepatic portosystemic shunt; ULN, upper limit of normal; WBC, white blood cells.

Financial support

The study was supported by an International Liver Scholarship program to TR by Gilead Sciences.

Conflict of interest

DC, MJ, LH, RP, PS, AFS declare no conflicts of interest. BeSi has received travel support from AbbVie and Gilead. DB has received travel support from AbbVie and Gilead, as well as speaker fees from AbbVie. BeSc received travel support from Abbvie, Gilead and Ipsen. MP is an investigator for Bayer, BMS, Lilly, and Roche; he received speaker honoraria from Bayer, BMS, Eisai, Lilly, and MSD; he is a consultant for Bayer, BMS, Ipsen, Eisai, Lilly, MSD, and Roche; he received travel support from Bayer and BMS. MT received grant support from Albireo, Cymabay, Falk, Gilead, Intercept, MSD, and Takeda, honoraria for consulting from BiomX, Boehringer Ingelheim, Falk, Genfit, Gilead, Intercept, Janssen, MSD, Novartis, Phenex, and Regulus, speaker fees from BMS, Falk, Gilead, Intercept and MSD, as well as travel support from Abbvie, Falk, Gilead and Intercept, MM served as a speaker and/ or consultant and/or advisory board member for AbbVie, Bristol-Myers Squibb, Gilead, Collective Acumen, and W. L. Gore & Associates and received travel support from AbbVie, Bristol-Myers Squibb, and Gilead. TR received grant support from Abbvie, Boehringer-Ingelheim, Gilead, MSD, Philips Healthcare, Gore; speaking honoraria from Abbvie, Gilead, Gore, Intercept, Roche, MSD; consulting/advisory board fee from Abbvie, Bayer, Boehringer-Ingelheim, Gilead, Intercept, MSD, Siemens; and travel support from Abbvie, Boehringer-Ingelheim, Gilead and Roche.

Please refer to the accompanying ICMJE disclosure forms for further details.

Authors' contributions

Study concept and design: DC, BeSi, MM, TR. Data acquisition: all authors

Analysis: DC, BeSi, MM, TR. Interpretation: all authors. Drafting the manuscript: DC, TR. Critical revision: BeSi, MJ, LH, DB, RP, PS, BeSc, AFS, MP, MT, MM

Data availability

Data are available upon reasonable request to the corresponding author.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jhep.2020.10.003.

References

Author names in bold designate shared co-first authorship

- D'Amico G, Garcia-Tsao G, Pagliaro L. Natural history and prognostic indicators of survival in cirrhosis: a systematic review of 118 studies. J Hepatol 2006;44:217–231.
- [2] D'Amico G, Morabito A, D'Amico M, Pasta L, Malizia G, Rebora P, et al. Clinical states of cirrhosis and competing risks. J Hepatol 2018;68:563– 576.
- [3] Ripoll C, Groszmann R, Garcia-Tsao G, Grace N, Burroughs A, Planas R, et al. Hepatic venous pressure gradient predicts clinical decompensation in patients with compensated cirrhosis. Gastroenterology 2007;133:481–488.
- [4] Arroyo V, Moreau R, Jalan R, Ginès P. Acute-on-chronic liver failure: a new syndrome that will re-classify cirrhosis. J Hepatol 2015;62:S131–S143.
- [5] Clària J, Stauber R, Coenraad M, Moreau R, Jalan R, Pavesi M, et al. Systemic inflammation in decompensated cirrhosis: characterization and role in acute-on-chronic liver failure. Hepatology 2016;64:1249–1264.
- [6] Garcia-Tsao G, Friedman S, Iredale J, Pinzani M. Now there are many (stages) where before there was one: in search of a pathophysiological classification of cirrhosis. Hepatology 2010;51:1445–1449.
- [7] de Franchis R. Evolving consensus in portal hypertension report of the Baveno IV consensus workshop on methodology of diagnosis and therapy in portal hypertension. J Hepatol 2005;43:167–176.
- [8] Garcia-Tsao G, Abraldes J, Berzigotti A, Bosch J. Portal hypertensive bleeding in cirrhosis: risk stratification, diagnosis and management – 2016 practice guidance by the American Association for the Study of Liver Diseases. Hepatology 2017;65:310–335.
- [9] Child CG, Turcotte JG. Surgery and portal hypertension. Major Probl Clin Surg 1964;1:1–85.
- [10] Kamath PS, Wiesner RH, Malinchoc M, Kremers W, Therneau TM, Kosberg CL, et al. A model to predict survival in patients with endstage liver disease. Hepatology 2001;33:464–470.
- [11] Durand F, Valla D. Assessment of the prognosis of cirrhosis: Child-Pugh versus MELD. J Hepatol 2005;42:S100–S107.
- [12] Groszmann RJ, Garcia-Tsao G, Bosch J, Grace ND, Burroughs AK, Planas R, et al. Beta-blockers to prevent gastroesophageal varices in patients with cirrhosis. N Engl J Med 2005;353:2254–2261.
- [13] Turco L, Garcia-Tsao G, Magnani I, Bianchini M, Costetti M, Caporali C, et al. Cardiopulmonary hemodynamics and C-reactive protein as prognostic indicators in compensated and decompensated cirrhosis. J Hepatol 2018;68:949–958.
- [14] Yotti R, Ripoll C, Benito Y, Catalina MV, Elizaga J, Rincon D, et al. Left ventricular systolic function is associated with sympathetic nervous activity and markers of inflammation in cirrhosis. Hepatology 2017;65:2019–2030.
- [15] Mehta G, Gustot T, Mookerjee R, Garcia-Pagan J, Fallon M, Shah V, et al. Inflammation and portal hypertension – the undiscovered country. J Hepatol 2014;61:155–163.
- [16] Bernardi M, Moreau R, Angeli P, Schnabl B, Arroyo V. Mechanisms of decompensation and organ failure in cirrhosis: from peripheral arterial vasodilation to systemic inflammation hypothesis. J Hepatol 2015;63:1272–1284.
- [17] Simbrunner B, Mandorfer M, Trauner M, Reiberger T. Gut-liver axis signaling in portal hypertension. World J Gastroenterol 2019;25:5897– 5917.
- [18] Moore KP, Wong F, Gines P, Bernardi M, Ochs A, Salerno F, et al. The management of ascites in cirrhosis: report on the consensus conference of the International Ascites Club. Hepatology 2003;38:258–266.
- [19] de Franchis R. Expanding consensus in portal hypertension. J Hepatol 2015;63:743-752.
- [20] Reiberger T, Puspok A, Schoder M, Baumann-Durchschein F, Bucsics T, Datz C, et al. Austrian consensus guidelines on the management and treatment of portal hypertension (Billroth III). Wien Klin Wochenschr 2017;129(Suppl 3):135–158.

- [21] Ferlitsch A, Bota S, Paternostro R, Reiberger T, Mandorfer M, Heinisch B, et al. Evaluation of a new balloon occlusion catheter specifically designed for measurement of hepatic venous pressure gradient. Liver Int 2015;35:2115–2120.
- [22] Reiberger T, Schwabl P, Trauner M, Peck-Radosavljevic M, Mandorfer M. Measurement of the hepatic venous pressure gradient and transjugular liver biopsy. J Vis Exp 2020;160. https://doi.org/10.3791/58819.
- [23] Trebicka J, Fernandez J, Papp M, Caraceni P, Laleman W, Gambino C, et al. The PREDICT study uncovers three clinical courses of acutely decompensated cirrhosis that have distinct pathophysiology. J Hepatol 2020;73:842–854.
- [24] Wiest R, Das S, Cadelina G, Garcia-Tsao G, Milstien S, Groszmann RJ. Bacterial translocation in cirrhotic rats stimulates eNOS-derived NO production and impairs mesenteric vascular contractility. J Clin Invest 1999;104:1223–1233.
- [25] Albillos A. Increased lipopolysaccharide binding protein in cirrhotic patients with marked immune and hemodynamic derangement. Hepatology 2003;37:208–217.
- [26] D'Amico G, Pasta L, Morabito A, D'Amico M, Caltagirone M, Malizia G, et al. Competing risks and prognostic stages in cirrhosis: a 25-year inception cohort study of 494 patients. Aliment Pharmacol Ther 2014;39:1180–1193.
- [27] Di Martino V, Coutris C, Cervoni JP, Dritsas S, Weil D, Richou C, et al. Prognostic value of C-reactive protein levels in patients with cirrhosis. Liver Transpl 2015;21:753–760.
- [28] Cervoni J, Thévenot T, Weil D, Muel E, Barbot O, Sheppard F, et al. Creactive protein predicts short-term mortality in patients with cirrhosis. J Hepatol 2012;56:1299–1304.
- [29] Schmidt-Arras D, Rose-John S. IL-6 pathway in the liver: from physiopathology to therapy. J Hepatol 2016;64:1403–1415.
- [30] D'Amico G, De Franchis R. Upper digestive bleeding in cirrhosis. Posttherapeutic outcome and prognostic indicators. Hepatology 2003;38:599–612.
- [31] Abraldes JG, Villanueva C, Banares R, Aracil C, Catalina MV, Garcia-Pagan JC. Hepatic venous pressure gradient and prognosis in patients with acute variceal bleeding treated with pharmacologic and endoscopic therapy. J Hepatol 2008;48:229–236.
- [32] Zipprich A, Garcia-Tsao G, Rogowsky S, Fleig WE, Seufferlein T, Dollinger MM. Prognostic indicators of survival in patients with compensated and decompensated cirrhosis. Liv Int 2012;32:1407–1414.
- [33] Jepsen P, Ott P, Andersen PK, Sorensen HT, Vilstrup H. The clinical course of alcoholic liver cirrhosis: a Danish population-based cohort study. Hepatology 2010;51:1675–1682.
- [34] Bernardi M, Angeli P, Claria J, Moreau R, Gines P, Jalan R, et al. Albumin in decompensated cirrhosis: new concepts and perspectives. Gut 2020;69:1127–1138.
- [35] Reiberger T, Mandorfer M. Beta adrenergic blockade and decompensated cirrhosis. J Hepatol 2017;66:849–859.
- [36] Mandorfer M, Reiberger T. Beta blockers and cirrhosis, 2016. Dig Liver Dis 2017;49:3–10.
- [37] Remmler J, Schneider C, Treuner-Kaueroff T. Increased level of interleukin 6 associates with increased 90-day and 1-year mortality in patients with end-stage liver disease. Clin Gastroenterol Hepatol 2018;16:730–737.
- [38] Mandorfer M, Schwabl P, Paternostro R, Pomej K, Bauer D, Thaler J, et al. Von Willebrand factor indicates bacterial translocation, inflammation, and procoagulant imbalance and predicts complications independently of portal hypertension severity. Aliment Pharmacol Ther 2018;47:980–988.
- [39] Fernández J, Clària J, Amorós A, Aguilar F, Castro M, Casulleras M, et al. Effects of albumin treatment on systemic and portal hemodynamics and systemic inflammation in patients with decompensated cirrhosis. Gastroenterology 2019;157:149–162.
- [40] Jalan R, Pavesi M, Saliba F, Amorós A, Fernandez J, Holland-Fischer P, et al. The CLIF Consortium Acute Decompensation score (CLIF-C ADs) for prognosis of hospitalised cirrhotic patients without acute-on-chronic liver failure. J Hepatol 2015;62:831–840.
- [41] Moreau R, Jalan R, Gines P, Pavesi M, Angeli P, Cordoba J, et al. Acute-onchronic liver failure is a distinct syndrome that develops in patients with acute decompensation of cirrhosis. Gastroenterology 2013;144:1426– 1437.